Author’s response to reviews

Title: Impact of an educational workshop on psychiatrists’ attitude towards psychosomatic medicine

Authors:
Franziska Baessler (franziska.baessler@med.uni-heidelberg.de)
Anja Ciprianidis (Anja.Ciprianidis@med.uni-heidelberg.de)
Fabienne Wagner (Fabienne.Wagner@med.uni-heidelberg.de)
Ali Zafar Rizvi (ali.zafar@med.uni-heidelberg.de)
Thanos Kanellopoulos (th.kanellopoulos@gmail.com)
Chiara Tabea Baumann (Tabea.Baumann@med.uni-heidelberg.de)
Clara-Louisa Müller (Clara-Louisa.Mueller@med.uni-heidelberg.de)
Jobst-Hendrik Schultz (Jobst-Hendrik.Schultz@med.uni-heidelberg.de)

Version: 4 Date: 23 Dec 2019

Author’s response to reviews:

1. Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. Thus, we request that you upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

The questionnaire has been uploaded as supplementary material.

2. Please note, if your manuscript is accepted you will not be able to make any changes to the authors, or order of authors, of your manuscript once the editor has accepted your manuscript for publication.

We would be grateful if the names of two authors can be modified from Fabienne Wagner to Fabienne Louise Wagner and Chiara Tabea Baumann to Tabea Chiara Baumann

3. Madhur Basnet: The reviewers have addressed the review comments well, have been well receptive and I have no further comments regarding the article.

Thanks a lot for the feedback and helping us to improve the article.
4. In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Psychiatry must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcpsychiatry.biomedcentral.com/).

We have added the Authors' Information section, which was missing in the earlier version.